Author's response to reviews

Title: Bilaterality can be ignored when analysing the revision risk of knee prostheses: A fraility analysis of 55 298 knee prostheses on 44 590 patients in the national Swedish Knee Arthroplasty Register Bilaterality can be ignored when analysing the revision risk of knee prostheses: A fraility analysis of 55 298 knee prostheses on 44 590 patients in the national Swedish Knee Arthroplasty Register

Authors:

Dr Otto Robertsson (otto.robertsson@ort.lu.se)
Jonas P Ranstam (jonas@ranstam.net)

Version: 2 Date: 3 Dec 2002

Besides a few changes in the text we have as Suggested by Assistant Editor Emma Veitch restructured the abstract and manuscript to better follow the guidelines of the BMC. Further, as recommended we now submit the article to BMC Musculoskeletal Disorders instead of BMC Medical Research Methodology.

Sincerely

Otto Robertsson